Dear Prof.(Reviewer #1),

Scientific Quality: Grade C (Good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Minor revision

Specific Comments to Authors:

1- The manuscript must be carefully proof-read for grammar and spelling issues.
The manuscript had already been re-edit by a professional English language editing company.

2- The molecular technique used in lines 203 and 214 should be mentioned in the manuscript.
It be mentioned in the manuscript.

3- It is recommended that authors change the numbering of the figures, A instead of 1A.
It be changed in the figures.

4- The authors must provide the requirements of ethics that their study was performed in accordance with local ethical guidelines.
The Research Ethics Committee of Rainbow Hospital of Xianyang approved this study (No. XYRR2016-026), and the children and their parents provided their written informed consent.

Thank you Prof,
Best regards,
Jinquan Wen